Author’s response to reviews

Title: Intakes of calcium, vitamin D, and dairy servings and dental plaque in older Danish adults

Authors:

Amanda RA Adegboye (aar@ipm.regionh.dk)
Lisa B Christensen (lbch@sund.ku.dk)
Poul Holm-Pedersen (holmp@sund.ku.dk)
Kirsten Avlund (kiav@sund.ku.dk)
Barbara J Boucher (bboucher@doctors.org.uk)
Berit L Heitmann (blh@ipm.regionh.dk)

Version: 4 Date: 11 March 2013

Author’s response to reviews: see over
Date: March 2013

Dear Editor-in-chief,

I am presenting for resubmission the revised manuscript entitled “INTAKES OF CALCIUM, VITAMIN D, AND DAIRY SERVINGS AND DENTAL PLAQUE IN OLDER DANISH ADULTS” for consideration for publication by the Nutrition Journal.

I have received the editor’s comments on the above paper and, as suggested, all reviewers’ comments have been incorporated in the revised version of the manuscript, which includes:

1. Inclusion of a comment about previous studies on the field of nutrition (in particular dairy intake) and oral health in the introduction section of the manuscript
2. Presentation of the full results, including the suggested table.

The changes are highlighted in the manuscript text. I hope that the changes are satisfactory and have made the manuscript acceptable for publication.

The results presented in this study have not been published, accepted, or submitted for publication elsewhere, nor does it duplicate already published material. The revised manuscript has been read and approved by all authors.

The authors declare no competing financial interests in relation to the work presented in the manuscript.

Yours sincerely,
Amanda Adegboye
Institute of Preventive Medicine
Frederiksberg Hospital, opg. 5,
Nordre Fasanvej 57,
2000 Frederiksberg, Copenhagen, Denmark